Author's response to reviews

Title: Health workforce responses to global health initiatives funding: a comparison of Malawi and Zambia

Authors:
- Ruairí Brugha (rbrugha@rcsi.ie)
- John Kadzandira (johnkadzandira@rcsi.ie)
- Joseph Simbaya (josephsimbaya@yahoo.com)
- Patrick Dicker (pdticker@rcsi.ie)
- Victor Mwapasa (vmwapasa@medcol.mw)
- Aisling Walsh (aislingwalsh@rcsi.ie)

Version: 6 Date: 4 August 2010

Author's response to reviews: see over
Health workforce responses to global health initiatives funding: a comparison of Malawi and Zambia

Formatting changes 04/08/10:

---------------------------
Major revisions (we require the author to make these changes)

Tables: we notice that you have included the tables as additional files. If you want the tables to be visible within the final published manuscript please include them in the manuscript in a tables section following the references. Please note that it is acceptable for tables in the manuscript to be in landscape format. The table title should appear above the table and should be only one sentence long. Please place any other text as a legend (or key), which should appear below the table. Alternatively, if you would like to keep the tables as additional files, please cite the files in the main manuscript text as Additional file 1 etc., and include an additional files section in the manuscript.

*Tables 1 to 3 are now included in the main manuscript following the references, in landscape format. The table titles are above the tables and each is one sentence long. Legends/keys are below the tables.*

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Abstract: Please remove the word count from the abstract

*This has been removed.*

Placement text: Please remove all figure and table placement text

*These have been removed.*

References: Please add the following details to reference 31: 'Capacity Project, ICS Integrare' to complete the publishing information.

*Added.*

Tables: Please remove the visible vertical lines from the tables

*Visible vertical lines have been removed.*

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

*The manuscript has been checked.*